Author's response to reviews

Title: Effects of antibody to receptor activator of nuclear factor kappa-B ligand on inflammation and cartilage degradation in collagen antibody-induced arthritis in mice

Authors:

Sakie Funato (sfunato@dent.showa-u.ac.jp)
Akihiro Matsunaga (aki_m168@med.showa-u.ac.jp)
Koei Oh (koei0525@med.showa-u.ac.jp)
Yoichi Miyamoto (yoichim@dent.showa-u.ac.jp)
Kentaro Yoshimura (kyoshimura@dent.showa-u.ac.jp)
Junichi Tanaka (itanka@dent.showa-u.ac.jp)
Dai Suzuki (dai.suzuki@dent.showa-u.ac.jp)
Risa Uyama (uyama@dent.showa-u.ac.jp)
Hiroaki Suzuki (suzuki.hc@csp.or.jp)
Kenji Mishima (mishima-k@dent.showa-u.ac.jp)
Masanori Nakamura (mas-nakamu@med.showa-u.ac.jp)
Osamu Namiki (o-namiki@db3.sho-net.ne.jp)
Kazuyoshi Baba (kazuyoshi@dent.showa-u.ac.jp)
Katsunori Inagaki (katsu@med.showa-u.ac.jp)
Ryutaro Kamijo (kamijor@dent.showa-u.ac.jp)

Version: 3
Date: 20 November 2014

Author's response to reviews:

November 20, 2014

Dr. Dirk Mielenz

The JNRBM Editorial Team

MS: 1931141132141149

Dear Dr. Dirk Mielenz,

Thank you very much for pointing out our mistake in Figure 4.
We have revised Figure 4 by replacing “RA-” with “RA-/Ab+”.

In addition, color of the text in Supplementary Information (Word file) has been changed from red to black. We would be happy if you could accept this revision.

With best regards,

Yoichi Miyamoto, Ph.D.

Department of Biochemistry, Showa University School of Dentistry
1-5-8 Hatanodai, Shinagawa, Tokyo 142-8555, Japan
E-mail: yoichim@dent.showa-u.ac.jp
Tel: 81-3-3784-8163
Fax: 81-3-3784-5555